Author's response to reviews

Title: Movements disorders in neuroleptic-naive patients with schizophrenia spectrum disorders

Authors: Moges Ayehu (ayehumoges@yahoo.com)
Teshome Shibre (teshome.kelkile@gmail.com)
Barkot Milkias (baromilki@gmail.com)
Abebaw Fekadu (abe.wassie@kcl.ac.uk)

Version: 5 Date: 25 September 2014

Author's response to reviews:

The Editor
BMC Psychiatry

Dear Sir/Dear Madam,

Re: Response to reviewers comments

MS-1081749867114088. Movement disorders in neuroleptic-naive patients with schizophrenia spectrum disorders in a low income setting. Moges Ayehu, Teshome Shibre, Barkot Milkias and Abebaw Fekadu

I am sorry about the error. The figure on prevalence of SMD is removed and the previous figure on nutritional status of patients is included again. We have also revised the manuscript to make it in line with the requirements of BMC Psychiatry.

We are grateful that the paper has been accepted for publication and for the editorial support.

Yours sincerely,

Abebaw

Dr Abebaw Fekadu, on behalf of the authors